Check for updates

OPEN ACCESS

EDITED BY Thomas H. Hampton, Dartmouth College, United States

REVIEWED BY Lauren Hatfield, Manchester Metropolitan University, United Kingdom Helen Gavillet, Manchester Metropolitan University, United Kingdom

*CORRESPONDENCE Ludovic Malvaux ⊠ ludovic.x.malvaux@gsk.com

[†]These authors have contributed equally to this work and share first authorship

SPECIALTY SECTION

This article was submitted to Infectious Agents and Disease, a section of the journal Frontiers in Microbiology

RECEIVED 14 November 2022 ACCEPTED 29 December 2022 PUBLISHED 24 February 2023

CITATION

Schoonbroodt S, Ichanté J-L, Boffé S, Devos N, Devaster J-M, Taddei L, Rondini S, Arora AK, Pascal T and Malvaux L (2023) Real-time PCR has advantages over culture-based methods in identifying major airway bacterial pathogens in chronic obstructive pulmonary disease: Results from three clinical studies in Europe and North America. *Front. Microbiol.* 13:1098133.

doi: 10.3389/fmicb.2022.1098133

COPYRIGHT

Malvaux, Schoonbroodt, Ichanté, Boffé, Devos, Devaster, Taddei, Rondini, Arora and Pascal, © 2023 GSK. This is an open-access article distributed under the terms of the Creative Commons Attribution License (CC BY). The use, distribution or reproduction in other forums is permitted, provided the original author(s) and the copyright owner(s) are credited and that the original publication in this journal is cited, in accordance with accepted academic practice. No use, distribution or reproduction is permitted which does not comply with these terms. Real-time PCR has advantages over culture-based methods in identifying major airway bacterial pathogens in chronic obstructive pulmonary disease: Results from three clinical studies in Europe and North America

Sonia Schoonbroodt^{1†}, Jean-Laurent Ichanté^{1†}, Sophie Boffé², Nathalie Devos¹, Jeanne-Marie Devaster¹, Laura Taddei³, Simona Rondini³, Ashwani Kumar Arora³, Thierry Pascal² and Ludovic Malvaux²*

¹GSK, Rixensart, Belgium, ²GSK, Wavre, Belgium, ³GSK, Siena, Italy

Introduction: We compared the performance of real-time PCR with culturebased methods for identifying bacteria in sputum samples from patients with chronic obstructive pulmonary disease (COPD) in three studies.

Methods: This was an exploratory analysis of sputum samples collected during an observational study of 127 patients (AERIS; NCT01360398), phase 2 study of 145 patients (NTHI-004; NCT02075541), and phase 2b study of 606 patients (NTHI-MCAT-002; NCT03281876). Bacteria were identified by culture-based microbiological methods in local laboratories using fresh samples or by real-time PCR in a central laboratory using frozen samples. *Haemophilus influenzae* positivity with culture was differentiated from *H. haemolyticus* positivity by microarray analysis or PCR. The feasibility of bacterial detection by culture-based methods on previously frozen samples was also examined in the NTHI-004 study.

Results: Bacterial detection results from both culture-based and PCR assays were available from 2,293 samples from AERIS, 974 from the NTHI-004 study, and 1736 from the NTHI-MCAT-002 study. Quantitative real-time PCR (qPCR) showed higher positivity rates than culture for *H. influenzae* (percentages for each study: 43.4% versus 26.2%, 47.1% versus 23.6%, 32.7% versus 10.4%) and Moraxella catarrhalis (12.9% versus 6.3%, 19.0% versus 6.0%, 15.5% versus 4.1%). In the NTHI-004 and NTHI-MCAT-002 studies, positivity rates were higher with qPCR for Streptococcus pneumoniae (15.6% versus 6.1%, 15.5% versus 3.8%); in AERIS, a lower rate with qPCR than with culture (11.0% versus 17.4%) was explained by misidentification of S. pseudopneumoniae/mitis isolates via conventional microbiological methods. Concordance analysis showed lowest overall agreement for H. influenzae (82.0%, 75.6%, 77.6%), due mainly to culture-negative/qPCR-positive samples, indicating lower sensitivity of the culture-based methods. The lowest positive agreement (culture-positive/ qPCR-positive samples) was observed for S. pneumoniae (35.1%, 71.2%, 71.2%). Bacterial load values for each species showed a proportion of culturenegative samples with a load detected by qPCR; for some samples, the loads were in line with those observed in culture-positive samples. In the NTHI-004 study, of fresh samples that tested culture-positive, less than 50% remained culture-positive when tested from freeze/thawed samples. In the NTHI-004 study, of fresh samples that tested culture-positive, less than 50% remained culture-positive when tested from freeze/thawed samples.

Discussion: Real-time PCR on frozen sputum samples has enhanced sensitivity and specificity over culture-based methods, supporting its use for the identification of common respiratory bacterial species in patients with COPD.

KEYWORDS

bacterial identification, culture, PCR, sensitivity, *Haemophilus influenzae, Moraxella catarrhalis, Streptococcus pneumoniae*, chronic obstructive pulmonary disease

Introduction

Chronic obstructive pulmonary disease (COPD) is a common condition characterized by persistent respiratory symptoms, often punctuated by acute exacerbations that can lead to hospitalization and a faster decline in lung function (Criner et al., 2015; Global Initiative for Chronic Obstructive Lung Disease, 2021). Different studies, including the observational Acute Exacerbation and Respiratory InfectionS in COPD (AERIS) study, have shown an association between exacerbation state and increased prevalence of airway bacteria, most commonly non-typeable Haemophilus influenzae (NTHi) and Moraxella catarrhalis, with some reports of an association with Streptococcus pneumoniae also (Garcha et al., 2012; Molyneaux et al., 2013; Huang et al., 2014; Millares et al., 2014; Wang et al., 2015; Wilkinson et al., 2017; Jubinville et al., 2018; Mayhew et al., 2018; Weeks et al., 2021). In the AERIS study, changes in the yearly COPD exacerbation rate were associated with changes in H. influenzae colonization (Wilkinson et al., 2019a), and nearly all (99%) H. influenzae isolates were non-typeable (Wilkinson et al., 2017).

A multi-component investigational vaccine was developed to reduce the frequency of acute exacerbations of COPD (AECOPD) associated with NTHi (Leroux-Roels et al., 2016), followed by a second related vaccine (NTHi-Mcat vaccine) to prevent NTHi-and *M. catarrhalis*-associated exacerbations (Van Damme et al., 2019). Both vaccines contain three conserved surface NTHi proteins, while the NTHi-Mcat vaccine also contains a surface protein from *M. catarrhalis* (Leroux-Roels et al., 2016; Van Damme et al., 2019). A phase 2 study of the NTHi vaccine (NTHI-004) and phase 2b study of the NTHi-Mcat vaccine (NTHI-MCAT-002) showed immunogenicity and no safety concerns when either vaccine was administered to patients with COPD (Wilkinson et al., 2019b; Andreas et al., 2022). In the NTHI-MCAT-002 study, while vaccination did not reduce the frequency of moderate/severe exacerbations, with no difference between groups in rate of AECOPD associated with NTHi or *M. catarrhalis*, observations suggested possible reductions in severe exacerbations and related hospitalizations in the vaccinated group versus placebo (Andreas et al., 2022).

In the AERIS, NTHI-004, and NTHI-MCAT-002 studies, sputum samples were collected from patients at regular intervals and at exacerbation (Wilkinson et al., 2017, 2019b; Andreas et al., 2022). Bacteria in freshly collected sputum were identified in local laboratories using conventional culture-based microbiological methods that were not harmonized. In parallel, frozen dithiothreitol (DTT)-treated sputum samples were transported to a central laboratory and analyzed by real-time PCR assay. Various studies have shown that molecular techniques, such as PCR, have better specificity and sensitivity than conventional culturedependent methods in the detection of airway bacteria in patients with COPD (Eser et al., 2012; Garcha et al., 2012; Bafadhel et al., 2015; Gadsby et al., 2015; Wilkinson et al., 2017). Also, culturebased methods require fresh samples and their reliability can be affected by antibiotic treatment (Oliver, 2010; Wu et al., 2014) as well as the microbiologist's skills in phenotypic identification methods. In contrast, a PCR assay can be used on frozen samples in a central location where it has been well characterized, thus avoiding the risk of variations in microbiological methods among individual local laboratories.

We now report on the performance of the microbiological assays used in the detection of bacteria in sputum samples in the AERIS, NTHI-004, and NTHI-MCAT-002 studies (Wilkinson et al., 2017, 2019b; Andreas et al., 2022). Quantitative and qualitative *H. influenzae, M. catarrhalis*, and *S. pneumoniae*

Abbreviations: AECOPD, acute exacerbations of chronic obstructive pulmonary disease; AERIS, Acute Exacerbation and Respiratory InfectionS in COPD; *algD*, GDP mannose dehydrogenase encoding gene; *CDS* 23, coding sequence 23 gene; *clfA*, clumping factor A encoding gene; *copB*, copB outer membrane protein encoding gene; COPD, chronic obstructive pulmonary disease; DTT, dithiothreitol; *lgtC*, lipo-oligosaccharide glycosyltransferase encoding gene; LOD, limit of detection; *lytA*, autolysin encoding gene; Mcat, *Moraxella catarrhalis*; NTHi, non-typeable *Haemophilus influenzae*; PCR, Polymerase Chain Reaction; qPCR, quantitative real-time PCR; STGG, skim milk-tryptone-glucose-glycerol.



identification results obtained with real-time PCR are compared with those obtained with culture-based methods. Additionally, using data from the NTHI-004 study, we examine the feasibility of freezing sputum samples before bacterial detection by culture in a central laboratory. Qualitative *Pseudomonas aeruginosa*, *Staphylococcus aureus*, and *Streptococcus pyogenes* identification data by PCR assay are also assessed from the three studies.

Figure 1 provides a plain language summary of the findings from these assessments.

Materials and methods

Sputum sample collection and processing

The sputum samples were collected in three clinical studies of adults with COPD: the prospective observational cohort study, AERIS, of 127 patients (NCT01360398; Wilkinson et al., 2017), the phase 2 placebo-controlled NTHI-004 study of an investigational NTHi vaccine in 145 patients (NCT02075541; Wilkinson et al., 2019b), and the phase 2b placebo-controlled NTHI-MCAT-002 study of an investigational NTHi-Mcat vaccine in 606 patients (NCT03281876; Andreas et al., 2022). The AERIS study was conducted in the United Kingdom between June 2011 and June 2014, the NTHI-004 study in the UK and Sweden between July 2014 and April 2017, and the NTHI-MCAT-002 study in Belgium, Canada, France, Germany, Italy, Spain, United Kingdom, and United States between November 2017 and March 2020. Study summaries are available¹ (study identifiers 114378, 200157, and 207489) and methods and results on primary and secondary endpoints of the studies have been published (Wilkinson et al., 2017; Mayhew et al., 2018; Osman et al., 2018; Wilkinson et al., 2019b; Malvisi et al., 2021; Andreas et al., 2022).

Sputum samples were obtained by spontaneous expectoration or induced, as per the investigator's judgement, at regular intervals during each study and at each exacerbation visit, and were processed according to standard methods, as described previously (Wilkinson et al., 2017, 2019b; Andreas et al., 2022). An acute exacerbation was defined as worsening of at least two major symptoms (dyspnea, sputum volume, and sputum purulence) or worsening of at least one major symptom and one minor symptom

¹ www.gsk-studyregister.com

(sore throat, cold symptoms, fever, increased cough, and increased wheeze) for at least two consecutive days.

Culture-based bacteriological methods and confirmatory assays

Freshly collected sputum samples were treated with DTT at 49 local microbiological laboratories associated with the clinical sites (one site for AERIS study, 15 for NTHI-004 study, and 67 for NTHI-MCAT-002 study). Bacterial species were identified by standard culture-based identification procedures (Murray et al., 2007), as summarized in Figure 2, and according to each local laboratory's routine methods, with minor adaptations (relating to, for example, the agar plate used for culture, sputum DTT pre-treatment, and semi-quantitative culture data reporting). Identification protocols included phenotypic characterization, semi-quantitative culture, and other methods such as Gram staining and biochemical methods. In the AERIS and NTHI-MCAT-002 studies, qualitative real-time PCR was used to further discriminate bacterial isolates of H. influenzae from H. haemolyticus by assessing the presence of two genes: P6 (outer membrane protein gene), which is conserved among H. influenzae and H. haemolyticus, and lgtC (lipo-oligosaccharide glycosyltransferase gene), which is ubiquitous in all H. influenzae but only 2% prevalent in *H. haemolyticus* (McCrea et al., 2008; Sandstedt et al., 2008; Abdeldaim et al., 2009). In the NTHI-004 study, culture plate sweeps were taken from presumptive H. influenzae-positive sputum in order to confirm H. influenzae positivity. This was part of an extended investigation of sputum sweep samples by Senti-HI molecular serotyping microarray analysis (data not shown) at BUGS Bioscience laboratory (London, United Kingdom). Culture plate sweeps from presumptive H. influenzae-positive sputum were stored in skim milk-tryptoneglucose-glycerol (STGG) medium (O'Brien et al., 2001; Kaijalainen et al., 2004) and genomic DNA extracted. Confirmation of species



Moraxella catarrhalis, Streptococcus pneumoniae, Pseudomonas aeruginosa, Staphylococcus aureus, and Streptococcus pyogenes. * Samples confirmed as *H. influenzae* positive after differentiation from other *Haemophilus* species (e.g., *H. haemolyticus*) by (NTHI-004 study) Senti-HI microarray analysis on stored *H. influenzae* sweep or (AERIS and NTHI-MCAT-002 studies) *lgtC/P6* qualitative real-time PCR assay. *algD*, GDP mannose dehydrogenase encoding gene; *CDS* 23, coding sequence 23; *clfA*, clumping factor A encoding gene; *copB*, copB outer membrane protein encoding gene; DTT, dithiothreitol; h, hours; *lgtC*, lipo-oligosaccharide glycosyltransferase encoding gene; *lytA*, autolysin encoding gene; STGG, skim milk-tryptone-glycerol medium. was based on the detection of species-specific gene targets for *H. influenzae*, in parallel with no detection of species-specific gene targets for eight other relevant *Haemophilus* species.

In the NTHI-004 study, a fraction of all DTT-treated sputum samples was stored in STGG medium before freezing and was shipped to the central laboratory for bacterial identification using standard culture-based methods.

Bacterial identification data generated using culture-based methods were qualitative (positive/negative). The bacterial load of *H. influenzae, M. catarrhalis,* and *S. pneumoniae* in cultured samples was estimated using a semi-quantitative quadrant assessment method (Murdoch et al., 2017) adapted according to local laboratory practices, using a '0, Few, 1+, 2+, 3+' scale, with '0' indicating absence and '3+' corresponding to the highest load (bacterial presence in at least three of four quadrants of the agar plate). The 'Few' category, which applied when only few, sparsely spread colonies were observed in any quadrant of the plate, was not recorded in the NTHI-004 study, according to the routine practices of the local laboratories.

PCR identification methods

Frozen aliquots of DTT-treated sputum samples were transported to the testing laboratory and analyzed by real-time PCR assay (Figure 2). The methods used for nucleic acid extraction and PCR are summarized in Table 1 and primers and probes sequences are provided in Supplementary Table 1.

Two triplex real-time PCR assays were employed that were developed by GSK and characterized in the AERIS study by DDL Diagnostic Laboratory (Rijswijk, the Netherlands) and in the NTHI-004 and NTHI-MCAT-002 studies by the central GSK laboratory (Wavre, Belgium). Following total nucleic acid extraction (see Table 1), a quantitative real-time PCR (qPCR) assay (Andreas et al., 2022) amplified DNA fragments of *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae* using the TaqMan Fast Advanced Master Mix kit (Life Technologies) on Viia7 or QuantStudio 7 equipment (Life Technologies). A second PCR assay was a qualitative assay that amplified *P. aeruginosa*, *S. aureus*, and *S. pyogenes* DNA using the same nucleic acids and the same PCR reagents and equipment (Life Technologies). The protocol for the second assay was essentially the same as for the first, with adaptation of the primers and probes sequences (Supplementary Table 1).

For the qPCR assay, three sets of primers and probes were designed from the conserved region of *lgtC* for *H. influenzae* (McCrea et al., 2008), the outer membrane protein *copB* gene for *M. catarrhalis* (Greiner et al., 2003), and the autolysin A gene (*lytA*) for *S. pneumoniae*. The *lytA* primers and probe sequences corresponded to those used in the lytA-CDC assay (Carvalho et al., 2007). The presence of *S. pyogenes, S. aureus,* and *P. aeruginosa* was determined using a qualitative real-time triplex PCR assay targeting conserved regions of the coding sequence 23 gene (*CDS23*), the clumping factor A encoding gene (*clfA*), and the GDP mannose dehydrogenase encoding gene (*algD*), respectively. *S. pyogenes* was not detected in any of the sputum samples.

PCR identification results were quantitative (copies/ml) or qualitative (positive/negative). Only qualitative data are available for P. aeruginosa and S. aureus. The sample was considered positive when the measured load was equal to or above the assay cut-off corresponding to the limit of detection (LOD) of the assay. LODs, expressed in copies/ml of DTT-treated sputum, were defined during characterization of the technical performance of the qPCR assay. Over time, there was re-assessment of some assay parameters, including LOD and quantitation limits, due to improvement of the methods (e.g., updated methods for nucleic acids extraction). There were consequent differences in LOD values between studies (see Table 1). The concentration of H. influenzae, M. catarrhalis, and S. pneumoniae DNA (bacterial load, copies/ml) in each sample was inferred from the calibration curve made from serial dilutions of a plasmid containing the sequences targeted by the assay and converted from copies/qPCR to copies/ml of DTT-treated sputum samples.

The specificity of these assays was verified theoretically (i.e., bioinformatic analyses) and experimentally (on related and unrelated bacteria and viruses commonly found in respiratory samples). No significant signal above the assays positivity cut-offs was observed and the sequencing of the PCR products generated from sputum samples confirmed that amplified material corresponded accurately to the reference sequences of the targeted pathogens.

Statistical analyses

This technical comparison was an exploratory analysis of data from sputum samples taken during the AERIS study (Wilkinson et al., 2017), NTHI-004 study (Wilkinson et al., 2019b), and NTHI-MCAT-002 study (Andreas et al., 2022), regardless of sputum quality and per-protocol defined cohorts. Positivity rates and their 95% confidence intervals (CIs) were calculated for the results of each assay. Concordance and agreement analyses were conducted on the qualitative identification results from samples for which both culture-based and PCR assay results were available. Overall agreement, positive agreement, and negative agreement were calculated as described in the footnote to Table 2. Dissymmetry in the number of discordant samples was evaluated by McNemar's test. In the analysis of bacterial load, samples with qPCR values above 0 copies/ml were considered for median computation and were plotted as a function of semi-quantitative culture results.

Results

Sampling

The total number of sputum samples from which bacterial detection results were available from both culture-based and PCR assays were 2,293 sputum samples from 127 patients in the AERIS study (years 1 and 2), 974 sputum samples from the

TABLE 1 Methods used for the triplex real-time PCR assays.

| A. Triplex quantitative real-time PCR (qPCR) assay to identify <i>Haemophilus influenzae</i> , <i>Moraxella catal Streptococcus pneumoniae</i> | <i>rhalis,</i> and |
|--|--------------------|
| | |

| Study | AERIS | NTHI-004 | NTHI-MCAT-002 | | |
|--|---|------------------------------------|-------------------------------|--|--|
| Nucleic acid extraction kit and protocol | MagNA Pure 96 equipment | Kingfisher equipment (Life | MagNA Pure 96 equipment | | |
| | (Roche) | Technologies) | (Roche) | | |
| | MagNA Pure 96 DNA and Viral | LSI MagVet Universal Isolation kit | MagNA Pure 96 DNA and Viral | | |
| | NA small volume kit (viral NA | | NA large volume kit (pathogen | | |
| | universal protocol) | | universal 500 protocol) | | |
| Real-time qPCR reagent | Taqman Fast Advanced Multiplex Universal Master Mix (Life Technologies) | | | | |
| H. influenzae target gene | lgtC | | | | |
| M. catarrhalis target gene | copB | | | | |
| S. pneumoniae target gene | lytA | | | | |
| Sputum-DTT input volume | 200 µl | 100 µl | 500 µl | | |
| Nucleic acid eluate volume | 50 µl | 80 µl | 100 µl | | |
| Nucleic acid volume input in qPCR reaction | 2 µl | 2 µl | 2 µl | | |
| Copies/qPCR to copies/ml conversion factor | Copies/ml=Copies/qPCR * 125 | Copies/ml=Copies/qPCR * 400 | Copies/ml=Copies/qPCR * 667 | | |
| Positivity cut-off (assay LOD) | | | | | |
| H. influenzae | 2,000 copies/ml | 16,400 copies/ml | 1,561 copies/ml | | |
| M. catarrhalis | 15,000 copies/ml | 44,800 copies/ml | 927 copies/ml | | |
| S. pneumoniae | 12,875 copies/ml | 12,400 copies/ml | 1,161 copies/ml | | |
| Testing laboratory | DDL Diagnostic | GSK Vaccines | GSK Vaccines | | |
| | | | | | |

B. Triplex qualitative real-time PCR assay to identify *Pseudomonas aeruginosa*, *Staphylococcus aureus*, and *Streptococcus pyogenes*

| Study | AERIS | NTHI-004 | NTHI-MCAT-002 | | |
|---|---|--------------------------------|-------------------------------|--|--|
| Nucleic acid extraction kit and protocol | MagNA Pure 96 equipment | Kingfisher equipment (Life | MagNA Pure 96 equipment | | |
| | (Roche) | Technologies) | (Roche) | | |
| | MagNA Pure 96 DNA and Viral | LSI MagVet Universal Isolation | MagNA Pure 96 DNA and Viral | | |
| | NA small volume kit (viral NA | Kit | NA large volume kit (pathogen | | |
| | universal protocol) | | universal 500 protocol) | | |
| Real-time PCR reagent | Taqman Fast Advanced Multiplex Universal Master Mix (Life Technologies) | | | | |
| P. aeruginosa target gene | algD | | | | |
| S. aureus target gene | clfA | | | | |
| S. pyogenes target gene | CDS 23 | | | | |
| Sputum-DTT input volume | 200 µl | 100 µl | 500 µl | | |
| Nucleic acid eluate volume | 50 µl | 80 µl | 100 µl | | |
| Nucleic acid volume input in PCR reaction | 2 µl | 2 µl | 2 µl | | |
| Positivity cut-off (assay LOD)* | | | | | |
| P. aeruginosa | 16,625 copies/ml | 21,600 copies/ml | 768 copies/ml | | |
| S. aureus | 17,500 copies/ml | 20,000 copies/ml | 1,108 copies/ml | | |
| Testing laboratory | DDL Diagnostic | GSK Vaccines | GSK Vaccines | | |

copB, copB outer membrane protein encoding gene; DTT, dithiothreitol; lgtC, lipo-oligosaccharide glycosyltransferase encoding gene; LOD, limit of detection; lytA, autolysin encoding gene. algD, GDP mannose dehydrogenase encoding gene; CDS 23, coding sequence 23; clfA, clumping factor A encoding gene; DTT, dithiothreitol; LOD, limit of detection. *S. pyogenes not detected in sputum samples.

NTHI-004 study (488 from 73 patients who received the investigational NTHi vaccine and 486 from 72 patients in the placebo group), and 1736 sputum samples from the

NTHI-MCAT-002 study (878 from 304 patients who received the NTHi-Mcat vaccine and 858 from 302 patients who received placebo).

| | Number of samples | | | Overall | Positive | Negative | P | | |
|--|-----------------------|---|---|---|---|-------------------|-------------------------------|-------|--------------------|
| Pathogen | Study | Culture- negative, PCR- negative | Culture- negative, PCR- positive | Culture- positive, PCR- negative | Culture- positive, PCR- positive | agreement" (%) | agreement [®] (%) | (%) | value ^a |
| Haemophilus influenzae ^e | AERIS | 1,173 | 367 | 9 | 538 | 81.98 | 98.35 | 76.17 | < 0.0001 |
| | NTHI- 004 | 501 | 229 | 5 | 223 | 75.57 | 97.81 | 68.63 | <0.0001 |
| | NTHI- MCAT- 002 | 1,140 | 379 | 1 | 176 | 77.59 | 99.44 | 75.05 | <0.0001 |
| Moraxella catarrhalis | AERIS | 1,824 | 143 | 4 | 128 | 93.00 | 96.97 | 92.73 | < 0.0001 |
| | NTHI- 004 | 773 | 127 | 2 | 55 | 86.52 | 96.49 | 85.89 | <0.0001 |
| | NTHI- MCAT- 002 | 1,462 | 201 | 4 | 68 | 88.18 | 94.44 | 87.91 | <0.0001 |
| Streptococcus pneumoniae | AERIS | 1,631 | 103 | 237 | 128 | 83.80 | 35.07 | 94.06 | < 0.0001 |
| | NTHI- 004 | 790 | 108 | 17 | 42 | 86.94 | 71.19 | 87.97 | <0.0001 |
| | NTHI- MCAT- 002 | 1,445 | 221 | 19 | 47 | 86.14 | 71.21 | 86.73 | <0.0001 |
| Pseudomonas aeruginosa | AERIS | 1,967 | 35 | 10 | 113 | 97.88 | 91.87 | 98.25 | 0.0002 |
| | NTHI- 004 | 918 | 13 | 4 | 21 | 98.22 | 84.00 | 98.60 | 0.0490 |
| | NTHI- MCAT- 002 | 1,593 | 49 | 8 | 85 | 96.71 | 91.40 | 97.02 | <0.0001 |
| Staphylococcus aureus | AERIS | 1,993 | 23 | 33 | 76 | 97.36 | 69.72 | 98.86 | 0.2288 |
| | NTHI- 004 | 874 | 46 | 10 | 26 | 94.14 | 72.22 | 95.00 | <0.0001 |
| | NTHI- MCAT- 002 | 1,517 | 108 | 26 | 85 | 92.28 | 76.58 | 93.35 | <0.0001 |

TABLE 2 Concordance analysis between bacterial pathogen identification results obtained by sputum culture and PCR assay (samples with both culture and PCR results).

^a Calculated as [(Culture-positive, PCR-positive) + (Culture-negative, PCR-negative)] divided by [(Culture-positive, PCR-positive) + (Culture-negative, PCR-negative) + (Culture-negative, PCR-negative) + (Culture-negative, PCR-negative) + (Culture-positive, PCR-negative) + (Culture-positive, PCR-negative) + (Culture-negative, PCR-negative)]. ^c Calculated as (Culture-negative, PCR-negative) + (Culture-negative, PCR-negative)]. ^d Difference between culture-negative/PCR-positive samples and culture-positive/PCR-negative samples: McNemar's test. ^c Samples confirmed as positive after differentiation from H. haemolyticus by Senti-HI microarray analysis or lgtC/P6 real-time PCR assay.

Bacterial identification

For *H. influenzae* and *M. catarrhalis* in all three studies, and *S. pneumoniae* in the NTHI-004 and NTHI-MCAT-002 studies, concordance analysis showed overall agreement of culture and qPCR data was between 75 and 93% (Table 2). Overall agreements were mainly impacted by the low negative agreement. Indeed, more samples were positive when identification was performed using qPCR, with statistically

significant (p < 0.0001) differences between positivity rates for samples assessed by culture versus those assessed by qPCR (Figure 3). In the concordance analysis, culture-negative/qPCRpositive results were associated with significant *p*-values (p < 0.0001; McNemar's test), indicating increased sensitivity with qPCR compared to culture-based methods for the three bacterial species (Table 2).

The concordance analysis also showed, for *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae*, few culture-positive and



PCR-positive sputum samples derived from frozen samples and assessed centrally (light brown bars). 95% CI, 95% confidence interval; n, number

of samples for which there were both valid culture and PCR results (POS/NEG) for each targeted pathogen.

qPCR-negative samples, apart from for *S. pneumoniae* in the AERIS study (Table 2). Moreover, the positivity rate for *S. pneumoniae* in the AERIS study was significantly higher (p < 0.0001) with culture identification than with qPCR (Figure 3). This discrepancy between culture and qPCR assay results observed with *S. pneumoniae* in the AERIS study was investigated further *via* various molecular techniques and mass spectrometry (see Supplementary material). This confirmed that the high rate of false-positive samples by culture-based methods was due to misidentification of samples containing *Streptococcus pseudopneumoniae* or *Streptococcus mitis* species.

For *P. aeruginosa*, the positivity rate was significantly higher (p = 0.0049) with PCR than with culture in the NTHI-MCAT-002 study only, while for *S. aureus*, a statistically significant difference was only observed in the NTHI-004 study (p = 0.0002) and the NTHI-MCAT-002 study (p < 0.0001), with higher positivity rates with PCR (Figure 3). Concordance analysis showed overall agreement of culture and PCR data was between 92% and 98% (Table 2). The difference between culture-negative/PCR-positive samples and culture-positive/PCR-negative samples was statistically significant (p < 0.05; McNemar's test) in all three

studies, apart from the discordant results for *S. aureus* in the AERIS study (Table 2).

The NTHI-004 study also evaluated the feasibility of freezing sputum samples before bacterial detection by culture in a central laboratory. The comparison of two culture-based methods for identifying *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae* showed lower positivity rates for STGG-frozen samples processed centrally than for fresh samples tested locally (Supplementary Table 2). Of fresh samples that tested *H. influenzae*-positive by culture, thawed frozen samples remained positive for 39.5% of samples and positive agreement was 47.1% for *M. catarrhalis* and 18.5% for *S. pneumoniae* (Supplementary Table 3).

Bacterial load

Bacterial load data generated by qPCR were plotted against the semi-quantitative results from culture-based methods for *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae*. Figure 4 shows data for samples with qPCR loads above >0 copy/reaction and associated with a semi-quantitative culture result. There was a



FIGURE 4

Box and dot plots of bacterial load results by quantitative real-time PCR (qPCR) as a function of semi-quantitative culture results. Median, first and third quartiles, and minimum and maximum data shown. ^a Culture-positive samples confirmed by Senti-HI microarray or lgtC/P6 real-time PCR assay. ^b Samples with qPCR loads above >0 copy/reaction and associated with a semi-quantitative culture result. See Table 1 for qPCR positivity cut-offs for H. influenzae, M. catarrhalis, and S. pneumoniae. Bacterial load in cultured samples estimated using the quadrant assessment method, according to a '0, Few, 1+, 2+, 3+' scale, with '0' indicating absence and '3+' corresponding to the highest load. 'Few' category not recorded in NTHI-004 study.

positive trend between the qPCR load and semi-quantitative culture results. For each species, the median load calculated by qPCR in sputum samples was higher in culture-positive samples than in culture-negative samples (Figure 4). Most culture-negative samples were associated with null bacterial loads by qPCR (not shown in Figure 4). For the AERIS study, of 1,540, 1,967, and 1,734 samples that were culture-negative for H. influenzae, M. catarrhalis, and S. pneumoniae, respectively, 1,017 (66.0%),

1,678 (85.3%), and 1,556 (89.7%) had null bacterial loads by qPCR. For the NTHI-004 study, of 730, 900, and 898 samples that were culture-negative for H. influenzae, M. catarrhalis, and S. pneumoniae, respectively, 463 (63.4%), 733 (81.4%), and 769 (85.6%) had null bacterial loads by qPCR. Similarly, for the NTHI-MCAT-002 study, of 1,519, 1,663, and 1,666 culture-negative samples for H. influenzae, M. catarrhalis, and S. pneumoniae, respectively, 1,015 (66.8%), 1,370 (82.4%), and 1,354 (81.3%) had

null bacterial loads by qPCR. Therefore, across all three studies, 33–37% (*H. influenzae*), 15–19% (*M. catarrhalis*), and 10–19% (*S. pneumoniae*) of culture-negative samples were detected by qPCR assay (below or above the positivity cut-off), with loads for some samples in line with those observed in culture-positive samples (Figure 4).

Discussion

We compared the performance of culture-based and PCR assays in identifying bacterial airway pathogens in sputum samples taken in three clinical studies of patients with COPD. In each study, two triplex real-time PCR assays were used, developed by GSK: a qPCR assay to detect and quantify *H. influenzae*, *M. catarrhalis*, and *S. pneumoniae* from sputum total nucleic acids, involving hydrolysis probes to maximize specificity, and a qualitative PCR assay to detect *P. aeruginosa*, *S. aureus*, and *S. pyogenes*. *S. pyogenes* was not identified in sputum samples from any of the studies. The three studies were conducted over the period 2011 to 2020, during which time the PCR assays were improved, such as in extraction methods for nucleic acids, leading to adjustments in LOD.

Our results show that real-time qPCR has higher specificity and sensitivity than culture-based methods for the detection of the most frequent bacterial species identified: H. influenzae, M. catarrhalis, and S. pneumoniae. The bacterial load analyses for these three species showed that some samples negative by culture contained a significant amount (over 10 million copies/ml) of bacterial DNA. Moreover, for P. aeruginosa and S. aureus, analysis of over 1,700 samples in the NTHI-MCAT-002 study showed significantly higher positivity rates with PCR than with culture-based detection methods for both species, although in the other two studies the difference was only significant for S. aureus in the NTHI-004 study. Overall, however, these data confirm previous reports of higher sensitivity with the PCR assay than with culture-based methods not only in COPD but also other conditions, such as otitis media and cystic fibrosis (Eser et al., 2012; Garcha et al., 2012; Bafadhel et al., 2015; Gadsby et al., 2015; Wilkinson et al., 2017; Gavillet et al., 2022). For example, one study of the detection of P. aeruginosa and S. aureus in cystic fibrosis found significantly lower detection of both lung pathogens by culture, which often did not detect either pathogen despite being found repeatedly by qPCR (Gavillet et al., 2022). In our study, qPCR positivity rates were around two-to four-fold higher for *H. influenzae* and *M. catarrhalis* in all three studies and for S. pneumoniae in the NTHI-004 and NTHI-MCAT-002 studies. In the concordance analysis, the lowest percentage positive agreement was for S. pneumoniae in the AERIS study (35%), reflecting the proportion of qPCRnegative samples among the culture-positive ones. Further analysis of this discrepant result led to the re-identification of a significant number of isolates, initially attributed to be *S. pneumoniae* isolates, as *S. pseudopneumoniae* or *S. mitis* by molecular techniques. The reason for this observation in the AERIS study may be related to characteristics of locally circulating strains or the specificity of the culture methods used to identify *S. pneumoniae* in the AERIS study laboratory; positive agreement in the other two studies was 71%. This suggests the culture-based method may have, in some settings, low specificity for *S. pneumoniae*.

A lack of specificity with culture-based methods is also demonstrated by the need to discriminate H. influenzae from H. haemolyticus isolates in presumptive H. influenzae-positive samples by using molecular methods (Murphy et al., 2007), as confirmed by qPCR or microarray in the three studies analyzed. Of the bacterial species assessed, the concordance analysis showed the lowest percentage overall agreement (76-82%) for H. influenzae (confirmed samples), due mainly to samples that were negative with culture-based detection but positive with qPCR. However, the presence of culture-negative samples associated with high bacterial DNA loads suggests this may have been partly due to the presence of non-culturable but viable bacteria (Oliver, 2010) or poor sensitivity of the culture method used. For example, use of bacitracin agar increases identification rates for H. influenzae (Harris et al., 2017) but this was not used in all laboratories. There is also a possibility that, although patients were instructed not to take an antibiotic before site visits, antibiotic treatment before sputum collection could have had an impact on the bacterial culture results (Brown and Spiegel, 2019). In this case, bacteria may not have been capable of growing on an agar plate but would have been detected by real-time PCR.

Our data suggest important advantages related to the specificity and sensitivity of real-time PCR over conventional culture-based methods for the assessment of airway bacteria in patients with COPD. Molecular assays can be performed on frozen sputum samples (Zhao et al., 2011; Cuthbertson et al., 2014), whereas culture needs to be done shortly after sputum collection to ensure sample integrity and bacteria viability (Baron et al., 2013). Frozen samples can be processed in a central laboratory with a well characterized PCR-based method, while fresh samples processed via conventional microbiological methods may require multiple local laboratories to avoid loss of viability. These local laboratories may not use exactly the same methods (see Figure 2), affecting the consistency of results (Hogardt et al., 2009). Culture of freeze/thawed samples in a central laboratory was examined in the NTHI-004 study as an option to avoid unharmonized methods across local laboratories. This analysis of the impact of freezing sputum samples before species identification by culture-based methods used STGG as storage medium since preserved H. influenzae, M. catarrhalis, and S. pneumoniae isolate viability had been reported with its use (Kaijalainen et al., 2004). However, the culture-positivity rate in STGG-frozen sputum samples was only around 50% or less of that in fresh samples, showing this is not a feasible option for evaluating sputum samples, thus providing further support for PCR testing of frozen samples in a central laboratory. The ability to freeze sputum samples is a particular advantage in phase 3 clinical trials of COPD patients, which generally require large numbers of sites to enroll a sufficient number of patients (Tashkin et al., 2008; Wedzicha et al., 2008; Calverley et al., 2009; Albert et al., 2011; Wedzicha et al., 2013; Pascoe et al., 2016). However, central analysis of frozen samples is associated with additional shipment costs and there is a risk of inappropriate sample management during the transport or freezing procedure, although there is also a management risk associated with fresh samples, for example, if not processed within the appropriate time window. Additionally, antibiotic usage before sputum sample collection can have an impact on the reliability of bacterial data obtained by culture (Wu et al., 2014; Mammen and Sethi, 2016; Jacobs et al., 2018), and the identification of pathogens by culture can be complicated by species overgrowth and contamination by commensal bacteria, while PCR allows direct species detection irrespective of these circumstances (Scoleri et al., 2016). PCR also has many advantages in terms of easy evaluation of bacterial load, it is relatively inexpensive, and it can be used to process more samples simultaneously than culture. These advantages assume the PCR assay has been appropriately designed (with highly specific oligonucleotide selection) and well characterized in terms of assay parameters, including LOD and quantitation limits, as was done for the PCR assays used in the AERIS, NTHI-004, and NTHI-MCAT-002 studies.

A major strength of these analyses is the large number (totaling 5,003) and multinational origin (Europe and North America) of sputum samples assessed for bacterial detection results from both culture-based and PCR assays. Also, the triplex real-time PCR assays used in each study were essentially the same, with any differences related to improvements made over time. The results of these analyses are limited by the possibility of false-positives resulting from the detection of low amounts of DNA, associated with dead bacteria, by qPCR. However, this was mitigated in these studies by using a LOD for each qPCR target as positivity cut-off (i.e., a sample was considered positive for a pathogen by qPCR if the observed load is equal to or above the corresponding LOD). We found many culture-and PCR-negative samples had PCR signals detected below the positivity cut-off, suggesting that using a PCR positivity cut-off is meaningful to limit the proportion of false-positives. Although qPCR can detect both viable and non-viable bacteria (Rudi et al., 2005), we found the bacterial loads measured by qPCR generally mirrored those measured by culture. Nevertheless, it would have been of interest to determine if samples found negative with culture-based techniques but positive on qPCR assay contained viable bacteria that were not culturable, as reported in other studies (Stenfors and Räisänen, 1992; Oliver, 2010; Lee and Bae, 2018), as well as the rate of non-viable bacteria present in samples recorded as qPCR-positive.

In conclusion, these results encourage the use of real-time PCR assays for the identification of respiratory bacteria in patients with COPD. PCR assay addresses some of the limitations of conventional culture-based methods in terms of specificity and sensitivity in detecting bacterial infection. PCR has additional advantages, including the ability to be performed in a centralized location on frozen samples and the capacity to detect viable but non-culturable bacteria. This supports the use of well characterized molecular methods for the identification and quantification of bacteria in future studies of patients with COPD, especially when working in a multicenter setting requiring sample testing in multiple laboratories.

Data availability statement

The datasets presented in this article are not readily available because anonymized individual participant data and study documents can be requested for further research from www. clinicalstudydatarequest.com.

Ethics statement

The studies involving human participants were reviewed and approved by the South West Hampshire Research Ethics Committee (Research Ethics Committee reference number: 11/ H0502/9) for the AERIS study; the Southampton and South West Hampshire Research Ethics Committee, United Kingdom, and Regional Research Ethics Committee, Gothenburg, Sweden for the NTHI-004 study; the ethics committee of each participating center for the NTHI-MCAT-002 study. The patients/participants provided their written informed consent to participate in this study.

Author contributions

SS, ND, J-MD, LT, SR, AKA, TP, and LM were involved in analysis conception and design. SB, ND, J-MD, LT, SR, AKA, TP, and LM were involved in acquisition and generation of data. All authors performed the data analysis and/or data interpretation. ND and LM provided materials for the analyses. All authors contributed to the article and approved the submitted version.

Funding

GlaxoSmithKline Biologicals SA funded the AERIS, NTHI-004, and NTHI-MCAT-002 clinical studies and was involved in all stages of study conduct, including analysis of the data. GlaxoSmithKline Biologicals SA also took in charge all costs associated with the development and publication of this manuscript.

Acknowledgments

The authors thank all participants, all members of the Clinical Laboratory Sciences teams involved in the generation of the data, and all external investigators and laboratory personal who generated the microbiological data from the studies AERIS (NCT01360398), NTHI-004 (NCT02075541), and NTHI-MCAT-002 (NCT03281876). We thank Jason Hinds (St George's, University of London, and BUGS Bioscience) for the development of the Senti-HI microarray tool and testing of the NTHI-004 study samples, and for permission to use the Senti-HI microarray data. We also thank Marie-Cécile Mortier (GSK) for her bioinformatics support. The authors also thank Business & Decision Life Sciences platform for editorial assistance, manuscript coordination, and writing support, on behalf of GSK. Joanne Knowles (independent medical writer, on behalf of GSK) provided medical writing support.

References

Abdeldaim, G. M., Strålin, K., Kirsebom, L. A., Olcén, P., Blomberg, J., and Herrmann, B. (2009). Detection of Haemophilus influenzae in respiratory secretions from pneumonia patients by quantitative real-time polymerase chain reaction. *Diagn. Microbiol. Infect. Dis.* 64, 366–373. doi: 10.1016/j.diagmicrobio.2009.03.030

Albert, R. K., Connett, J., Bailey, W. C., Casaburi, R., Cooper, J. A. Jr., Criner, G. J., et al. (2011). Azithromycin for prevention of exacerbations of COPD. *N. Engl. J. Med.* 365, 689–698. doi: 10.1056/NEJMoa1104623

Andreas, S., Testa, M., Boyer, L., Brusselle, G., Janssens, W., Kerwin, E., et al. (2022). Non-typeable Haemophilus influenzae-Moraxella catarrhalis vaccine for the prevention of exacerbations in chronic obstructive pulmonary disease: a multicentre, randomised, placebo-controlled, observer-blinded, proof-of-concept, phase 2b trial. *Lancet Respir. Med.* 10, 435–446. doi: 10.1016/S2213-2600(21)00502-6

Bafadhel, M., Haldar, K., Barker, B., Patel, H., Mistry, V., Barer, M. R., et al. (2015). Airway bacteria measured by quantitative polymerase chain reaction and culture in patients with stable COPD: relationship with neutrophilic airway inflammation, exacerbation frequency, and lung function. *Int. J. Chron. Obstruct. Pulmon. Dis.* 10, 1075–1083. doi: 10.2147/copd.s80091

Baron, E. J., Miller, J. M., Weinstein, M. P., Richter, S. S., Gilligan, P. H., Thomson, R. B. Jr., et al. (2013). A guide to utilization of the microbiology laboratory for diagnosis of infectious diseases: 2013 recommendations by the Infectious Diseases Society of America (IDSA) and the American Society for Microbiology (ASM). *Clin. Infect. Dis.* 57, e22–e121. doi: 10.1093/cid/cit278

Brown, R. M., and Spiegel, R. (2019). Antibiotic prophylaxis in patients with COPD. Am. Fam. Physician 99

Calverley, P. M., Rabe, K. F., Goehring, U. M., Kristiansen, S., Fabbri, L. M., and Martinez, F. J. (2009). Roflumilast in symptomatic chronic obstructive pulmonary disease: two randomised clinical trials. *Lancet* 374, 685–694. doi: 10.1016/s0140-6736(09)61255-1

Carvalho, M. G. S., Tondella, M. L., McCaustland, K., Weidlich, L., McGee, L., Mayer, L. W., et al. (2007). Evaluation and improvement of real-time PCR assays targeting lytA, ply, and psaA genes for detection of pneumococcal DNA. *J. Clin. Microbiol.* 45, 2460–2466. doi: 10.1128/jcm.02498-06

Criner, G. J., Bourbeau, J., Diekemper, R. L., Ouellette, D. R., Goodridge, D., Hernandez, P., et al. (2015). Prevention of acute exacerbations of COPD: American College of Chest Physicians and Canadian Thoracic Society Guideline. *Chest* 147, 894–942. doi: 10.1378/chest.14-1676

Cuthbertson, L., Rogers, G. B., Walker, A. W., Oliver, A., Hafiz, T., Hoffman, L. R., et al. (2014). Time between collection and storage significantly influences bacterial

Conflict of interest

SS, J-LI, SB, ND, J-MD, LT, SR, AKA, TP, and LM are employed by GSK. SS, ND, J-MD, LT, TP, and LM hold shares in GSK. J-MD is also a designated inventor on patents owned by GSK.

Publisher's note

All claims expressed in this article are solely those of the authors and do not necessarily represent those of their affiliated organizations, or those of the publisher, the editors and the reviewers. Any product that may be evaluated in this article, or claim that may be made by its manufacturer, is not guaranteed or endorsed by the publisher.

Supplementary material

The Supplementary material for this article can be found online at: https://www.frontiersin.org/articles/10.3389/fmicb.2022.1098133/ full#supplementary-material

sequence composition in sputum samples from cystic fibrosis respiratory infections. J. Clin. Microbiol. 52, 3011–3016. doi: 10.1128/jcm.00764-14

Eser, O. K., Alp, S., Ergin, A., Ipçi, K., Alp, A., Gür, D., et al. (2012). Comparison of culture and real-time PCR methods in the detection of Streptococcus pneumoniae and Haemophilus influenzae in acute otitis media effusion specimens. *Mikrobiyol. Bul.* 46, 676–681.

Gadsby, N. J., McHugh, M. P., Russell, C. D., Mark, H., Conway Morris, A., Laurenson, I. F., et al. (2015). Development of two real-time multiplex PCR assays for the detection and quantification of eight key bacterial pathogens in lower respiratory tract infections. *Clin. Microbiol. Infect.* 21, 788.e1–788.e13. doi: 10.1016/j. cmi.2015.05.004

Garcha, D. S., Thurston, S. J., Patel, A. R., Mackay, A. J., Goldring, J. J., Donaldson, G. C., et al. (2012). Changes in prevalence and load of airway bacteria using quantitative PCR in stable and exacerbated COPD. *Thorax* 67, 1075–1080. doi: 10.1136/thoraxjnl-2012-201924

Gavillet, H., Hatfield, L., Rivett, D., Jones, A., Maitra, A., Horsley, A., et al. (2022). Bacterial culture underestimates lung pathogen detection and infection status in cystic fibrosis. *Microbiol Spectr* 10:e0041922. doi: 10.1128/spectrum.00419-22

Global Initiative for Chronic Obstructive Lung Disease. (2021). Global strategy for the diagnosis, management, and prevention of chronic obstructive pulmonary disease. 2022 report [Online]. Available at: https://goldcopd.org/wp-content/uploads/2021/11/GOLD-REPORT-2022-v1.0-12Nov2021_WMV.pdf [Accessed Jun 30, 2022].

Greiner, O., Day, P. J., Altwegg, M., and Nadal, D. (2003). Quantitative detection of Moraxella catarrhalis in nasopharyngeal secretions by real-time PCR. *J. Clin. Microbiol.* 41, 1386–1390. doi: 10.1128/JCM.41.4.1386-1390.2003

Harris, T. M., Rumaseb, A., Beissbarth, J., Barzi, F., Leach, A. J., and Smith-Vaughan, H. C. (2017). Culture of non-typeable Haemophilus influenzae from the nasopharynx: not all media are equal. *J. Microbiol. Methods* 137, 3–5. doi: 10.1016/j.mimet.2017.03.012

Hogardt, M., Ulrich, J., Riehn-Kopp, H., and Tümmler, B. (2009). EuroCareCF quality assessment of diagnostic microbiology of cystic fibrosis isolates. J. Clin. Microbiol. 47, 3435–3438. doi: 10.1128/jcm.01182-09

Huang, Y. J., Sethi, S., Murphy, T., Nariya, S., Boushey, H. A., and Lynch, S. V. (2014). Airway microbiome dynamics in exacerbations of chronic obstructive pulmonary disease. *J. Clin. Microbiol.* 52, 2813–2823. doi: 10.1128/JCM.00035-14

Jacobs, D. M., Ochs-Balcom, H. M., Zhao, J., Murphy, T. F., and Sethi, S. (2018). Lower airway bacterial colonization patterns and species-specific interactions in chronic obstructive pulmonary disease. J. Clin. Microbiol. 56, e00330-e00318. doi: 10.1128/jcm.00330-18

Jubinville, E., Veillette, M., Milot, J., Maltais, F., Comeau, A. M., Levesque, R. C., et al. (2018). Exacerbation induces a microbiota shift in sputa of COPD patients. *PLoS One* 13:e0194355. doi: 10.1371/journal.pone.0194355

Kaijalainen, T., Ruokokoski, E., Ukkonen, P., and Herva, E. (2004). Survival of Streptococcus pneumoniae, Haemophilus influenzae, and Moraxella catarrhalis frozen in skim milk-tryptone-glucose-glycerol medium. *J. Clin. Microbiol.* 42, 412–414. doi: 10.1128/JCM.42.1.412-414.2004

Lee, S., and Bae, S. (2018). Molecular viability testing of viable but non-culturable bacteria induced by antibiotic exposure. *Microb. Biotechnol.* 11, 1008–1016. doi: 10.1111/1751-7915.13039

Leroux-Roels, G., Van Damme, P., Haazen, W., Shakib, S., Caubet, M., Aris, E., et al. (2016). Phase I, randomized, observer-blind, placebo-controlled studies to evaluate the safety, reactogenicity and immunogenicity of an investigational non-typeable Haemophilus influenzae (NTHi) protein vaccine in adults. *Vaccine* 34, 3156–3163. doi: 10.1016/j.vaccine.2016.04.051

Malvisi, L., Taddei, L., Yarraguntla, A., Wilkinson, T. M. A., Arora, A. K., and AERIS Study Group (2021). Sputum sample positivity for Haemophilus influenzae or Moraxella catarrhalis in acute exacerbations of chronic obstructive pulmonary disease: evaluation of association with positivity at earlier stable disease timepoints. *Respir. Res.* 22:67. doi: 10.1186/s12931-021-01653-8

Mammen, M. J., and Sethi, S. (2016). COPD and the microbiome. *Respirology* 21, 590–599. doi: 10.1111/resp.12732

Mayhew, D., Devos, N., Lambert, C., Brown, J. R., Clarke, S. C., Kim, V. L., et al. (2018). Longitudinal profiling of the lung microbiome in the AERIS study demonstrates repeatability of bacterial and eosinophilic COPD exacerbations. *Thorax* 73, 422–430. doi: 10.1136/thoraxjnl-2017-210408

McCrea, K. W., Xie, J., LaCross, N., Patel, M., Mukundan, D., Murphy, T. F., et al. (2008). Relationships of nontypeable Haemophilus influenzae strains to hemolytic and nonhemolytic Haemophilus haemolyticus strains. *J. Clin. Microbiol.* 46, 406–416. doi: 10.1128/jcm.01832-07

Millares, L., Ferrari, R., Gallego, M., Garcia-Nuñez, M., Pérez-Brocal, V., Espasa, M., et al. (2014). Bronchial microbiome of severe COPD patients colonised by Pseudomonas aeruginosa. *Eur. J. Clin. Microbiol. Infect. Dis.* 33, 1101–1111. doi: 10.1007/s10096-013-2044-0

Molyneaux, P. L., Mallia, P., Cox, M. J., Footitt, J., Willis-Owen, S. A., Homola, D., et al. (2013). Outgrowth of the bacterial airway microbiome after rhinovirus exacerbation of chronic obstructive pulmonary disease. *Am. J. Respir. Crit. Care Med.* 188, 1224–1231. doi: 10.1164/rccm.201302-03410C

Murdoch, D. R., Morpeth, S. C., Hammitt, L. L., Driscoll, A. J., Watson, N. L., Baggett, H. C., et al. (2017). Microscopic analysis and quality assessment of induced sputum from children with pneumonia in the PERCH study. *Clin. Infect. Dis.* 64, S271–S279. doi: 10.1093/cid/cix083

Murphy, T. F., Brauer, A. L., Sethi, S., Kilian, M., Cai, X., and Lesse, A. J. (2007). Haemophilus haemolyticus: a human respiratory tract commensal to be distinguished from Haemophilus influenzae. *J. Infect. Dis.* 195, 81–89. doi: 10.1086/509824

Murray, P. R., Baron, E. J., Jorgensen, J. H., Landry, M. L., and Pfaller, M. A. (2007). *Manual of clinical microbiology*, *9th Edn*. Washington, DC: American Society of Microbiology Press.

O'Brien, K. L., Bronsdon, M. A., Dagan, R., Yagupsky, P., Janco, J., Elliott, J., et al. (2001). Evaluation of a medium (STGG) for transport and optimal recovery of Streptococcus pneumoniae from nasopharyngeal secretions collected during field studies. *J. Clin. Microbiol.* 39, 1021–1024. doi: 10.1128/jcm.39.3.1021-1024.2001

Oliver, J. D. (2010). Recent findings on the viable but nonculturable state in pathogenic bacteria. *FEMS Microbiol. Rev.* 34, 415–425. doi: 10.1111/j.1574-6976.2009.00200.x

Osman, K. L., Jefferies, J. M. C., Woelk, C. H., Devos, N., Pascal, T. G., Mortier, M. C., et al. (2018). Patients with chronic obstructive pulmonary disease harbour a variation of Haemophilus species. *Sci. Rep.* 8:14734. doi: 10.1038/ s41598-018-32973-3 Pascoe, S. J., Lipson, D. A., Locantore, N., Barnacle, H., Brealey, N., Mohindra, R., et al. (2016). A phase III randomised controlled trial of single-dose triple therapy in COPD: the IMPACT protocol. *Eur. Respir. J.* 48, 320–330. doi: 10.1183/13993003.02165-2015

Rudi, K., Moen, B., Dromtorp, S. M., and Holck, A. L. (2005). Use of ethidium monoazide and PCR in combination for quantification of viable and dead cells in complex samples. *Appl. Environ. Microbiol.* 71, 1018–1024. doi: 10.1128/aem.71.2.1018-1024.2005

Sandstedt, S. A., Zhang, L., Patel, M., McCrea, K. W., Qin, Z., Marrs, C. F., et al. (2008). Comparison of laboratory-based and phylogenetic methods to distinguish between Haemophilus influenzae and H. haemolyticus. *J. Microbiol. Methods* 75, 369–371. doi: 10.1016/j.mimet.2008.06.023

Scoleri, G. P., Choo, J. M., Leong, L. E., Goddard, T. R., Shephard, L., Burr, L. D., et al. (2016). Culture-independent detection of nontuberculous mycobacteria in clinical respiratory samples. *J. Clin. Microbiol.* 54, 2395–2398. doi: 10.1128/jcm.01410-16

Stenfors, L. E., and Räisänen, S. (1992). Occurrence of Streptococcus pneumoniae and Haemophilus influenzae in otitis media with effusion. *Clin. Otolaryngol. Allied Sci.* 17, 195–199. doi: 10.1111/j.1365-2273.1992.tb01825.x

Tashkin, D. P., Celli, B., Senn, S., Burkhart, D., Kesten, S., Menjoge, S., et al. (2008). A 4-year trial of tiotropium in chronic obstructive pulmonary disease. *N. Engl. J. Med.* 359, 1543–1554. doi: 10.1056/NEJMoa0805800

Van Damme, P., Leroux-Roels, G., Vandermeulen, C., De Ryck, I., Tasciotti, A., Dozot, M., et al. (2019). Safety and immunogenicity of non-typeable Haemophilus influenzae-Moraxella catarrhalis vaccine. *Vaccine* 37, 3113–3122. doi: 10.1016/j. vaccine.2019.04.041

Wang, H., Gu, X., Weng, Y., Xu, T., Fu, Z., Peng, W., et al. (2015). Quantitative analysis of pathogens in the lower respiratory tract of patients with chronic obstructive pulmonary disease. *BMC Pulm. Med.* 15:94. doi: 10.1186/s12890-015-0094-z

Wedzicha, J. A., Calverley, P. M., Seemungal, T. A., Hagan, G., Ansari, Z., and Stockley, R. A. (2008). The prevention of chronic obstructive pulmonary disease exacerbations by salmeterol/fluticasone propionate or tiotropium bromide. *Am. J. Respir. Crit. Care Med.* 177, 19–26. doi: 10.1164/rccm.200707-973OC

Wedzicha, J. A., Decramer, M., Ficker, J. H., Niewoehner, D. E., Sandstrom, T., Taylor, A. F., et al. (2013). Analysis of chronic obstructive pulmonary disease exacerbations with the dual bronchodilator QVA149 compared with glycopyrronium and tiotropium (SPARK): a randomised, double-blind, parallel-group study. *Lancet Respir. Med.* 1, 199–209. doi: 10.1016/s2213-2600(13)70052-3

Weeks, J. R., Staples, K. J., Spalluto, C. M., Watson, A., and Wilkinson, T. M. A. (2021). The role of non-typeable Haemophilus influenzae biofilms in chronic obstructive pulmonary disease. *Front. Cell. Infect. Microbiol.* 11:720742. doi: 10.3389/fcimb.2021.720742

Wilkinson, T. M. A., Aris, E., Bourne, S., Clarke, S. C., Peeters, M., Pascal, T. G., et al. (2017). A prospective, observational cohort study of the seasonal dynamics of airway pathogens in the aetiology of exacerbations in COPD. *Thorax* 72, 919–927. doi: 10.1136/thoraxjnl-2016-209023

Wilkinson, T. M. A., Aris, E., Bourne, S. C., Clarke, S. C., Peeters, M., Pascal, T. G., et al. (2019a). Drivers of year-to-year variation in exacerbation frequency of COPD: analysis of the AERIS cohort. *ERJ Open Res* 5, 00248–02018. doi: 10.1183/23120541.00248-2018

Wilkinson, T. M. A., Schembri, S., Brightling, C., Diar Bakerly, N., Lewis, K., MacNee, W., et al. (2019b). Non-typeable Haemophilus influenzae protein vaccine in adults with COPD: a phase 2 clinical trial. *Vaccine* 37, 6102–6111. doi: 10.1016/j. vaccine.2019.07.100

Wu, S., Li, X., Gunawardana, M., Maguire, K., Guerrero-Given, D., Schaudinn, C., et al. (2014). Beta-lactam antibiotics stimulate biofilm formation in non-typeable Haemophilus influenzae by up-regulating carbohydrate metabolism. *PLoS One* 9:e99204. doi: 10.1371/journal.pone.0099204

Zhao, J., Li, J., Schloss, P. D., Kalikin, L. M., Raymond, T. A., Petrosino, J. F., et al. (2011). Effect of sample storage conditions on culture-independent bacterial community measures in cystic fibrosis sputum specimens. *J. Clin. Microbiol.* 49, 3717–3718. doi: 10.1128/jcm.01189-11